

Review 1. This is a practical message which will be of help to ER abdominal teams.

Reply: We agree with your comment and thank you very much.

Reviewer 2: the discussion is too long - rather than a review of prior cases, the discussion should focus more on the current case

Reply: We agree with your suggestion. We have deleted some of the content in the discussion, which is shown in red.